Author's response to reviews

Title: Thoracoscopic plication for a Huge Thoracic Meningocele in a Patient with Neurofibromatosis

Authors:

Heng-Chung Chen (96367@cch.org.tw)
Ping-Hsien Chang (144016@cch.org.tw)
Shang-Wun Jhang (buwang1@gmail.com)
Bing-Yen Wang (156283@cch.org.tw)

Version: 3 Date: 18 September 2013

Author's response to reviews: see over
Dear Editor,

Enclosed is our manuscript entitled “Thoracoscopic plication for a Huge Thoracic Meningocele in a Patient with Neurofibromatosis” by Wang et al, in the hope that you will consider publishing it. Intrathoracic meningocele with neurofibromatosis type I is a rare disease, and the optimal treatment is still unknown. Herein, we present the case of a 48-year-old female with a huge thoracic meningocele associated with cutaneous neurofibromatosis type I and kyphoscoliosis of the thoracic spine. The large thoracic meningocele was successfully treated through the thoracoscopic plication. The new method was never reported before.

No manuscript with substantially the same content has been submitted or published elsewhere. All authors are aware of and agree with the contents of the paper, and with their names being listed as the authors of the paper. Thank you very much.

Sincerely yours,

Bing-Yen Wang, MD

Division of Thoracic Surgery, Department of Surgery, Changhua Christian Hospital

No. 135, Nanxium St, Changhua City, Changhua country 500, Taiwan

Phone: 886-(2)-7238595
Fax: 886-(2)-7228289

E-mail: 156283@cch.org.tw